Author's response to reviews

Title: The feasibility of a randomised controlled trial of Dyadic Developmental Psychotherapy

Authors:

Fiona Turner-Halliday (fiona.turner@glasgow.ac.uk)
Nicholas Watson (nicholas.watson@glasgow.ac.uk)
Nicole Soldner (nsoldne2@gmail.com)
Kathleen Boyd (kathleen.boyd@glasgow.ac.uk)
Helen Minnis (helen.minnis@glasgow.ac.uk)

Version: 2 Date: 18 April 2013

Author's response to reviews:

Dear editorial committee,

As requested, we have inserted two new statements 1) In the methods section we have inserted a sentence to say that our local NHS ethics committee decided that the study did not require ethical approval; 2) that the methods and analysis that we adopted meet the RATS guidelines for qualitative research.

We hope this this will suffice, but please do not hesitate to contact me should anything else require amendment.

Many thanks,

Dr Fiona Turner-Halliday